Author’s response to reviews

Title: Stampidine Prevents Mortality In An Experimental Mouse Model Of Viral Hemorrhagic Fever Caused By Lassa Virus

Authors:

Fatih M Uckun (fatih_uckun@ih.org)
Alexander S Petkevich (asp@briem.ac.by)
Alexei O Vassilev (avassilev@ih.org)
Ms Heather E Tibbles (htibbles@ih.org)
Leonid P Titov (briem@hmti.ac.by)

Version: 2 Date: 28 Aug 2003

Revision for format